Author's response to reviews

Title: Comparing the Analgesic Effect of Heat Pack Containing Iron Chip and Ibuprofen for Primary Dysmenorrhea: A randomized controlled trial

Authors:

Shahin-Dokht Navvabi-Rigi MSc (shahinnavabi@yahoo.com)
Fatihe Kerman-Saravi MSc (fkermansaravi@yahoo.com)
Ali Navidian PhD (alinavidian@gmail.com)
Leila Safabakhsh MSc (lili_safa@yahoo.com)
Somaye Khazaian MSc (m_khazaian@yahoo.com)
Shahla Shafiee PhD (shafieesh@gmail.com)
Ameneh Safazadeh MSc (safarzadeh40@yahoo.com)
Tahmineh Salehian MSc (tahminehsalehian@yahoo.com)

Version: 2 Date: 15 August 2011

Author's response to reviews: see over
Trial registration number: IRCT201107187038N2

Consent to participate in study: Written consent for publication was obtained from the patient.

Competing interest: the authors declare that they have no competing interests.

Authors’ contributions
SHDNR conceived of the study, and participated in its design and coordination and helped to draft the manuscript.

FKS participated in the sequence alignment and drafted the manuscript.

AN participated in the design of the study and performed the statistical analysis.

LS participated in the Criticality revised the manuscripts.

AS and SKH and SHSH and TS were involved in data collection and provided and the primary draft manuscript and other authors read and approved the final manuscript. All authors read and approved the final manuscript.